Author's response to reviews

Title: Use of aspirin for primary and secondary prevention of cardiovascular disease in diabetic patients in an ambulatory care setting in Spain

Authors:

Antoni Sicras-Mainar (asicras@bsa.gs)
Ruth Navarro-Artieda (rnavarro.germanstrias@gencat.net)
Xavi Frias-Garrido (xfrias@bsa.gs)
Rafael Ruiz-Riera (rruiz@bsa.gs)
Javier Rejas-Gutierrez (javier.rejas@pfizer.com)
Jaime Fernandez-de-Bobadilla (jaime.fernandez@pfizer.com)

Version: 6 Date: 21 August 2007

Author's response to reviews:

The ethical considerations were including in the methods of the manuscript